Operability Report 

11/29/2006
Accelerator
Maintain 5 pass 3.303 GeV machine (326 MeV per linac)
· High current run for A and C lasers
· Weekly Mott measurements
· No beam to BSY dump
· Keep RF economized

Hall A

Elastic Electron Scattering Off 3He and 4He
· 5 pass 3.303 GeV
· Up to 100 uA

· No beam in Compton chicane

· FFB on

· ep or arc energy measurement as requested (procedure will be updated)
Hall B

G 13 Kaon Production on the Deuteron Using Polarized Photons
· 4 pass 2.649 GeV
· 1 to 40 nA
· Current and orbit locks on

· 60 Hz noise vs aperture steering

· Development of 60 Hz suppression tool in progress

Hall C

G0 Backwards Angle







· 1 pass 0.690 GeV
· Up to 60 uA

· Weekly Mott, Moeller, and arc energy measurements
· RF phase vs halo

· Position asymmetry distribution

· FFB on with energy
Items Needing Attention
· Online CS tool
Beam Transport Issues 

Accelerator
· 31 MHz Beam Quality Issues 
· 3 hall delivery causes BLM trips in hall A
Hall A

· Compton rates (signal vs. background)
· Several attempts at tuning were made.  

· There is some correlation between slit position and halo rates

· Remember the PID lock for the vertical position on target 
Hall B

· Current stability, 60 Hz noise 

· Optics and EGG staff helped reduce 60 Hz noise. Beam centered on hall B chopping slit, Path length adjusted, Dispersion suppressed from 5S through 6T line, 60 Hz suppression system tested.

Hall C






· FFB on (energy)
· Access taken to investigate alternate magnet locations

· Multiple BLM trips

· Adjusted launch, steering and cycled hall line correctors as per TS Guide
· Combined optics solution is still a possibility
· High halo on 10/31/06 caused by DC beam coming from the hall A laser.
· High halo on 11/2/06 caused by stray magnetic field after Moeller back out
· Adjusted quad at 3C16 to lower rates 
